Author's response to reviews

Title: A closer look at the increase in suicide rates in South Korea from 1986-2005

Authors:

Jin-Won Kwon (winor@chol.com)
Heeran Chun (heeranchun@hanmail.net)
Sung-il Cho (persontime@hotmail.com)

Version: 3 Date: 24 February 2009

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

Based on your review comments, we revised as below.

1. Reference 9: This is a book with institutional author and we changed as below.

2. Reference 22: We found one spelling error in URL and revised as below.
   Russian analytical digest No. 48 (17Oct2008) [http://www.res.ethz.ch/analysis/rad/]

3. Tables shading: We revised it bold.

4. Table size: We reduced the Table to fit the A4 portrait page.

5. Figure title and legends: We deleted the title, number and legend in Figures and legend was added in Manuscript.

6. Minor changes were revised as below
   Title page - Removed word count information.
   Abstract - Removed keywords from the manuscript and the submission system.
   Changed 'Contributors' to 'Authors' contributions'
   Changed 'Conflict of interests statement' to 'Competing interests'.
   Removed the 'Ethical approval' section.
   Table title and legend - A text as a legend (or key) was placed below the table.
   Figures – Changed Fig to Figure.

Thanks and best regards,
Sung-il Cho